
\documentclass[10pt]{amsart}
\usepackage[cp1251]{inputenc}
\usepackage[english,russian]{babel}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{graphicx}


\def\udcs{519.21}
\def\mscs{60F10, 60A99}
\setcounter{page}{144}


\theoremstyle{plain}
\newtheorem{lemma}{Lemma}
\newtheorem{theorem}{Theorem}
\newtheorem{definition}{Definition}
\newtheorem{corollary}{Corollary}
\newtheorem{proposition}{Proposition}
\theoremstyle{definition}
\newtheorem{remark}{Remark}
\newtheorem*{example*}{Example}

\def\logo{{\bf\huge S\raisebox{0.2ex}{\hspace{0.55ex}\raisebox{0.05ex}e\hspace{-1.65ex}$\bigcirc$}MR}}

\def\semrtop
     {
  \vbox{
     \noindent\logo
     \hspace{80mm}\raisebox{1ex}{ISSN 1813-3304 }

     \vspace{5mm}

     \begin{center}
     {\huge СИБИРСКИЕ \ ЭЛЕКТРОННЫЕ} \\[2mm]
     {\huge МАТЕМАТИЧЕСКИЕ ИЗВЕСТИЯ} \\[2mm]
     {\large Siberian Electronic Mathematical Reports} \\[1mm]
     {\LARGE\tt{http://semr.math.nsc.ru}}\\[0.5mm]
%     {\small Институт математики им. С.Л.Соболева СО РАН}\\[-1mm]
%     {\small Sobolev Institute of Mathematics SB RAS}
     \end{center}
     \vspace{-3mm}
     \noindent
     \begin{tabular}{c}
     \hphantom{aaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa} \\
     \hline\hline
     \end{tabular}

     \vspace{1mm}
     {\flushleft\it Том 16, стр. 144--144 (2019) \hspace{65mm}{\rm\small УДК \udcs}}
     \newline
     	{\rm\small DOI~10.33048/semi.2019.16.xxx}\hphantom{aaaaaaaaaaaaaaaaaaiaaaaaaaaaaaaa}{\rm\small MSC\ \ \mscs }
  }%\vbox
}

\begin{document}
\begin{thebibliography}{1}

\bibitem{bor}
A. A. Borovkov, {\it Asymptotic Analysis of Random Walks}, Cambridge University Press, 2020.
\bibitem{bor_mog}
A. A. Borovkov, A. A. Mogulskii, {\it Moderately Large Deviation Principles for the Trajectories of Random Walks and Processes with Independent Increments}, Theory of Probability \& Its Applications, \textbf{58}:4 (2014), 562--581.
\bibitem{lebedev}
A. A. Lebedev, B. A. Lebedev, {\it Generalization of the Expectation for a Unified Description of Random and Indefinite Factors}, Theory of Probability \& Its Applications, \textbf{38}:3 (1994), 470--478.
\bibitem{deuschel}
J.-D. Deuschel, D.W. Stroock, {\it Large deviations}, Academic Press, 1989.
\bibitem{dembo}
A. Dembo, O. Zeitouni. {\it Large deviations techniques and applications. 2nd ed.}, Springer, 1998.
\bibitem{peng}
S. Peng, {\it Nonlinear expectations and stochastic calculus under uncertainty}, Springer, 2019.
\bibitem{peng_g}
S. Peng, {\it G-Expectation, G-Brownian motion and related stochastic calculus of Ito’s type}, Stoch. Anal. Appl., \textbf{2} (2006), 541--567.
\bibitem{peng_mult}
S. Peng, {\it Multi-dimensional G-Brownian motion and related stochastic calculus under G-expectation}, Stoch. Process. Their Appl., \textbf{118} (2008), 2223--2253.
\bibitem{peng_lln_clt}
S. Peng, {\it Law of large numbers and central limit theorem under nonlinear expectations}, Probab. Uncertain. Quant. Risk., \textbf{4}:4 (2019).
\bibitem{chen}
Z. Chen, P. Wu, B. Li, {\it A strong law of large numbers for non-additive probabilities}, Int. J. Approx. Reasoning., \textbf{54}:3 (2013), 365--377.
\bibitem{liu}
W. Liu, Y. Zhang, {\it Large deviation principle for linear processes generated by real stationary sequences under the sublinear expectation}, Commun. Stat.-Theor. M., (2021).
\bibitem{hu}
F. Hu, {\it On Cram\'{e}r's theorem for capacities}, Comptes Rendus Mathematique., \textbf{348} (2010), 1009--1013.
\bibitem{gao_xu}
F. Gao, M. Xu, {\it Large deviations and moderate deviations for independent random variables under sublinear expectations}, Scientia Sinica., \textbf{41}:4 (2011), 337--352.
\bibitem{cao} X. Cao, {\it An upper bound of large deviations for capacities}, Math. Probl. Eng., (2014).
\bibitem{chen_feng} Z. J. Chen, Z. W. Feng, {\it Large deviation for negatively dependent random variables under sublinear expectation}, Commun. Stat.-Theor. M., \textbf{45}:2 (2016), 400--412.
\bibitem{tan_zong} Y. Z. Tan, G. F. Zong, {\it Large deviation principle for random variables under sublinear expectations on $\mathbb{R}^d$}, Math. Anal. Appl., \textbf{488} (2020).
\bibitem{chen_xiong} Z. Chen, J. Xiong, {\it Large deviation principle for diffusion processes under a sublinear expectation}, Sci. China Math, \textbf{55} (2012), 2205--2216.
\bibitem{gao_jiang} F. Gao, H. Jiang, {\it Large deviations for stochastic differential equations driven by G-Brownian motion}, Stochastic Processes Appl., \textbf{120}:11 (2010), 2212--2240.
\bibitem{log_zhou}
Q. Q. Zhou, A. V. Logachov, {\it Moderate deviations principle for independent random variables under sublinear expectations}, Sib. Elektron. Math. Rep., \textbf{18} (2021), 817--826.
\bibitem{guo}
S. Guo, Z. Yong, {\it Moderate deviation principle for m-dependent random variables under the sublinear expectation}, AIMS Mathematics, \textbf{7}:4 (2022), 5943--5956.

\end{thebibliography}

\renewcommand{\refname}{References}


\thispagestyle{empty}

\title[On the moderate deviation principle for $m$--dependent variables]{On the moderate deviation principle for $m$--dependent random variables with sublinear expectation}
\author{{E.V. EFREMOV, A.V. LOGACHOV}}
\address{Egor Vladimirovich Efremov
\newline\hphantom{iii} Novosibirsk State University,
\newline\hphantom{iii} 2, Pirogova str.,
\newline\hphantom{iii} Novosibirsk, 630090, Russia}
\email{e.efremov@g.nsu.ru}

\address{Artem Vasilhevich Logachov
\newline\hphantom{iii} Lab. of Probability Theory and Math. Statistics,
\newline\hphantom{iii} Sobolev Institute of Mathematics,
\newline\hphantom{iii} 4, Koptyuga ave.,
\newline\hphantom{iii} Novosibirsk, 630090, Russia
\newline\hphantom{iii} Dep. of Computer Science in Economics, Novosibirsk State Technical University
\newline\hphantom{iii} pr. K. Marksa, 20,
\newline\hphantom{iii} 630073, Novosibirsk, Russia}
\email{omboldovskaya@mail.ru}


\thanks{\sc Efremov, E.V., Logachov, A.V., On moderate deviation principle for $m$--dependent variables with sublinear expectation}
\thanks{\copyright \ 2023 Efremov E.V. Logachov, A.V.}
\thanks{\rm The work is supported by the Mathematical Center in Akademgorodok under the agreement No. 075-15-2022-282 with the Ministry of Science and Higher Education of the Russian Federation.}
\thanks{\it Поступила 1 января 2015 г., опубликована 31 декабря 2015 г.}%


\semrtop \vspace{1cm}
\maketitle {\small
\begin{quote}
\noindent{\sc Abstract. } In this paper, we obtain the moderate deviation principle for sums of $m$--dependent strictly stationary random variables in the space with sublinear expectation. Unlike known results, we will require random variables to satisfy a less restrictive Cramer--like condition.\medskip

\noindent{\bf Keywords:} large deviation principle, moderate deviation principle, sublinear expectation, $m$--dependent random variables, stationary sequences.
 \end{quote}
}

\selectlanguage{english}
\section{Introduction}

Since 1990s the theory of random variables in the sublinear expectation space has been actively developed.
This progress can be attributed to two main factors.
On the one hand, sublinear expectation retains most of the properties of the regular expectation, which makes it possible to transfer with some changes the main results of classical probability theory to it. 
On the other hand,
the utilization of sublinear expectation spaces allows for statistical inferences to be made even in situations where limited information is available about the assumed distribution of a random elements,  which makes this theory highly applicable in solving some practical problems.

Before giving definitions, we briefly review known results. Apparently, the first work in this fields is by Lebedev \cite{lebedev}, who investigated a special case of sublinear expectation, and gave examples of its applications.
Peng \cite{peng_g} introduced the notions of independent random elements, normal distribution and Brownian motion in sublinear expectation space, known as $G$--normal distribution and $G$--Brownian motion.
Moreover, Peng constructed an analogue of the stochastic integral and investigated both regular and backward stochastic differential equations.
In paper \cite{peng_mult} Peng extended results of \cite{peng_g} to multidimensional case.
Further advancements were made by Peng \cite{peng_lln_clt} and Chen, Wu, and Li  \cite{chen}, who derived analogues of the law of large numbers and central limit theorem for random variables in sublinear expectation space. Additionally, they obtained analogues of the main inequalities of classical probability theory for sublinear expectation space.
For more detailed information about recent results in the field of theory of sublinear expectation we refer to the book by Peng \cite{peng}.

\par
Now let us review the results that are directly related to the large deviation principle (LDP).
It is important to note that LDP, as well as its special case, the moderate deviation principle (MDP), are valuable tools for estimating the probabilities of rare events. LDP
is mostly related to rough exponential asymptotics of probabilities of rare events for sequences of random elements. More detailed information about LDP and MDP in classical probability theory can be found in books by Borovkov \cite{bor}, Deuschel and Stroock \cite{deuschel} and Dembo and Zeitouni \cite{dembo}.

In the context of sublinear expectation space, one of the earliest results related to LDP for random variables can be attributed to Hu \cite{hu}, who derived the upper bound of Cram\'{e}r's theorem for capacities.
Gao and Xu \cite{gao_xu} obtained LDP for independent and identically distributed random variables in sublinear expectation space. Cao \cite{cao} obtained an upper bound of LDP for independent and identically distributed $d$--dimensional random variables in sublinear expectation space. 

Chen and Feng \cite{chen_feng} investigated LDP for negatively dependent random variables in sublinear expectation space. They also obtained an upper bound for MDP. Tan and Zong \cite{tan_zong} obtained LDP for $d$--dimensional random variables in sublinear expectation space without requiring them to be independent and identically distributed. 

Works by Chen and Xiong \cite{chen_xiong} and by Gao and Jiang \cite{gao_jiang}
were concerned with LDP for solutions of stochastic differential equations,
in which the stochastic integral is constructed with $G$--Brownian motion.

Zhou and Logachov \cite{log_zhou} obtained MDP for weakly dependent random variables in sublinear expectation space, without requiring them to be identically distributed. Guo and Yong \cite{guo} obtained MDP for strictly stationary $m$--dependent random variables in sublinear expectation space.\par
In this paper we investigate MDP for strictly stationary $m$--dependent random variables with weaker moment condition than that used in \cite{log_zhou} and \cite{guo}. Thus, we generalize the results obtained in \cite{log_zhou} and \cite{guo}.
We also note that in case when random variables are independent and considered in classical probability space, the result of this paper directly follows from \cite{bor_mog}, (see also chapter 5 of \cite{bor}).\par

Now we review needed definitions and related symbols.
Let $(\Omega, \mathfrak{F})$ be a sample space and a $\sigma$--algebra of its subsets, respectively. We denote by $\mathcal{H}$ a linear space of real-valued functions (random variables) defined on $\Omega$ and measurable with respect to $\mathfrak{F}$, such that:

1) $c \in \mathcal{H}$ for any constant $c \in \mathbb{R}$;

2) if $\vert X\vert \in\mathcal{H}$ and $\vert Y\vert \leqslant \vert X\vert$, then $Y\in \mathcal{H}$.
\\Now we give definition of sublinear expectation defined on $\mathcal{H}$.
\begin{definition}
  Sublinear expectation on $\mathcal{H}$ is a functional $\mathbb{E} : \mathcal{H} \to \mathbb{R}$ that satisfies the following properties: for all $X, Y \in \mathcal{H}$
\begin{enumerate}
  \item[(i)] monotonicity: if $X\geqslant Y$, then $\mathbb{E} [X] \geqslant \mathbb{E} [Y]$;
  \item[(ii)] constant preserving: $\mathbb{E} [c] = c,\; \forall c \in \mathbb{R}$;
  \item[(iii)] sub--additivity: $\mathbb{E} [X+Y] \leqslant \mathbb{E} [X] + \mathbb{E} [Y]$;
  \item[(iv)] positive homogeneity: $\mathbb{E} [\lambda X] = \lambda \mathbb{E} [X],\; \forall \lambda \geq 0$.
\end{enumerate}
The triplet $(\Omega, \mathcal{H}, \mathbb{E})$ is called a sublinear expectation space.
\end{definition}
\begin{remark}[Peng \cite{peng}]\label{from_2_3}
  From properties (ii) and (iii) it follows that for any $X, Y \in \mathcal{H}$
  \begin{enumerate}
    \setcounter{enumi}{4}
    \item[(v)] $\mathbb{E}[X+c] =\mathbb{E}[X] + c,\; \forall c \in \mathbb{R}$;
    \item[(vi)] $\mathbb{E}[X-Y]\geqslant \mathbb{E}[X]-\mathbb{E}[Y]$.
  \end{enumerate}
\end{remark}
\begin{definition}
  Let $\textbf{X}_1$ and $\textbf{X}_2$ be two $n$--dimensional random vectors defined in sublinear expectation spaces $(\Omega_1, \mathcal{H}_1, \mathbb{E}_1)$ and $(\Omega_2, \mathcal{H}_2, \mathbb{E}_2)$ respectively. They are said to be identically distributed, denoted $\textbf{X}_1 \overset{d}{=} \textbf{X}_2$, if
  \[
    \mathbb{E}_1[\varphi(\textbf{X}_1)] = \mathbb{E}_2[\varphi(\textbf{X}_2)]
  \]
  for any Borel measurable function $\varphi$ on $\mathbb{R}^n$, such that $\varphi(\textbf{X}_1)\in \mathcal{H}$ and $\varphi(\textbf{X}_2)\in\mathcal{H}$.
\end{definition}

\begin{definition}
  Let $\{X_i\}_{i=1}^{n}$ be random variables in a sublinear expectation space $(\Omega, \mathcal{H}, \mathbb{E})$. Random variable $X_n$ is said to be independent from $X_1,\ldots,X_{n-1}$, if for any set of nonnegative Borel measurable functions $\varphi_i$ on $\mathbb{R}$, such that $\varphi_i(X_i)\in\mathcal{H}$, we have
  \begin{equation*}
    \mathbb{E} \left[\prod\limits_{i=1}^{n}\varphi_i(X_i)\right] = \mathbb{E} \left[\prod\limits_{i=1}^{n-1}\varphi_i(X_i)\right]\mathbb{E}[\varphi_n(X_n)].
  \end{equation*}
\end{definition}


\begin{definition}
  A sequence of random variables $\{X_i\}_{i=1}^{\infty}$ is called $m$--dependent if there exists $m\geqslant 1$ such that for every $n\geqslant 1$, every $j\geqslant m +1$ and every pair of nonnegative Borel measurable functions $\varphi_1$ and $\varphi_2$, such that $\varphi_1(X_1,\ldots,X_n)\in\mathcal{H}$ and $\varphi_2 (X_{n+m+1},\ldots,X_{n+j})\in\mathcal{H}$, we have
  \begin{multline*}
    \mathbb{E} \left[\varphi_1(X_1,\ldots,X_n)\varphi_2 (X_{n+m+1},\ldots,X_{n+j}) \right]\\
    = \mathbb{E} \left[\varphi_1(X_1,\ldots,X_n)\right] \mathbb{E} \left[\varphi_2 (X_{n+m+1},\ldots,X_{n+j})\right].
  \end{multline*}
\end{definition}

\begin{definition}
  A sequence of random variables $\{X_i\}_{i=1}^{\infty}$ on $(\Omega, \mathcal{H}, \mathbb{E})$ is called strictly stationary, if for every $n\geqslant 1$, every $k\geqslant 1$ and every Borel measurable function $\varphi$ on $\mathbb{R}^n$, such that $\varphi (X_1, X_2,\ldots, X_n)\in\mathcal{H}$ and $\varphi (X_{1+k}, X_{2+k},\ldots,X_{n+k})\in\mathcal{H}$, we have
  \[
    \mathbb{E} [\varphi (X_1, X_2,\ldots, X_n)] = \mathbb{E} [\varphi (X_{1+k}, X_{2+k},\ldots,X_{n+k})].
  \]
\end{definition}

\begin{definition}
  We define an upper probability (a capacity) $\mathbb{V}$ as follows
  \[
    \mathbb{V}(A) := \mathbb{E} [\mathbf{I}(A)],\quad A \in \mathfrak{F}.
  \]
\end{definition}

\begin{definition}\label{mdp_def}
  A sequence of random variables $s_n$ in $(\Omega, \mathcal{H}, \mathbb{E})$ is said to satisfy LDP in $\mathbb{R}$ with rate function $I$ and speed $1/\psi(n) : \lim\limits_{n\to\infty}\psi(n) = \infty$, if for any Borel set $B \in \mathfrak{B}(\mathbb{R})$
  \begin{equation}\label{mdp_def_upper}
    \limsup\limits_{n\to\infty} \frac{1}{\psi(n)}\ln \mathbb{V}(s_n \in B)\leqslant -\inf\limits_{y\in [B]}I(y),
  \end{equation}
  \begin{equation}\label{mdp_def_lower}
    \liminf\limits_{n\to\infty} \frac{1}{\psi(n)}\ln \mathbb{V}(s_n \in B)\geqslant -\inf\limits_{y\in (B)}I(y),
  \end{equation}
  where $[B]$ is the closure of set $B$, and $(B)$ is the interior of the set $B$. Here we put $I(\varnothing) := \infty$.
\end{definition}
\begin{definition}\label{mdp_def2}
  A sequence of random variables $s_n$ in $(\Omega, \mathcal{H}, \mathbb{E})$ is said to satisfy MDP if it satisfies LDP with the same parameters as LDP for a sequence of independent and identically distributed Gaussian random variables with zero mean in classical theory.
\end{definition}

We will consider a sequence of strictly stationary $m$--dependent random variables $\{X_i\}_{i=1}^{\infty}$ in a sublinear expectation space $(\Omega, \mathcal{H}, \mathbb{E})$ with
\begin{equation}\label{centered}
  \mathbb{E} [X_i] = \mathbb{E} [-X_i] = 0,\quad i \in \mathbb{N},
\end{equation}
and
\begin{equation}\label{homogen}
  \lim\limits_{n\to\infty}\frac{1}{n}\mathbb{E} \left[\sum\limits_{i=1}^{n}X_i \right]^2 = \sigma^2 < \infty.
\end{equation}
In addition, we require the following moment condition to be satisfied. For some $q>0, \alpha \in(0,1], M > 0$\par
\begin{equation}\label{exp_moment}
  \mathbb{E} [e^{q\vert X_1\vert^{\alpha}}]< M.
\end{equation}\par
Let $x=x(n)$ be a sequence of positive real numbers, such that
\begin{equation}\label{seq}
  \lim\limits_{n\to\infty} \frac{x(n)}{\sqrt{n}} = \infty,\quad \lim\limits_{n\to\infty}\frac{x^{2-\alpha}(n)}{n} = 0.
\end{equation}
Denote
\[
  S_k := \sum\limits_{i=1}^{k}X_i,\quad k\in \mathbb{N}.
\]
\par
The following theorem is the main result of the paper.
\begin{theorem}[\textbf{MDP for sums of $m$--dependent r.v.s}]\label{mdp_theorem}
  Let conditions (\ref{centered})-- (\ref{seq}) be satisfied. Then the sequence
  \[
    s_n = \frac{S_n}{x(n)}
  \]
  satisfies LDP in $\mathbb{R}$ with speed $1/\psi(n) := \frac{n}{x^2(n)}$ and rate function $I(y) := \frac{y^2}{2\sigma^2}$, $y \in \mathbb{R}$.
\end{theorem}

Consider an example.
\begin{example*}
  Let $(\Omega, \mathfrak{F})$ be a measurable space, and let $\mathbf{P}_1$ and $\mathbf{P}_2$ be a pair of probability measures defined on that space. We denote expectations corresponding to $\mathbf{P}_1$ and $\mathbf{P}_2$ as $\mathbf{E}_1$ and $\mathbf{E}_2$ respectively. Let $\mathcal{H}$ be a linear space of all real-valued functions $X$ defined on $\Omega$, such that they are measurable with respect to $\mathfrak{F}$, and
  \[
    \max\limits_{i\in \{1,2\}}\mathbf{E}_i\vert X\vert < \infty.
  \]
  We define
  \[
    \mathbb{E} [X] = \max\limits_{i\in \{1,2\}}\mathbf{E}_i X,
  \]
  for $X\in\mathcal{H}$. The triplet $(\Omega, \mathcal{H}, \mathbb{E})$ is a sublinear expectation space.\par
  Let $\{X_i\}_{i=1}^{\infty}$ be a sequence of independent random variables, such that every random variable from a sequence $\{X_{2i + 1}\}_{i=0}^{\infty}$ has absolutely continuous probability distribution with pdf
  \[
    f_{X}(x) = \frac{1}{2}\frac{\beta}{\theta}\left(\frac{\vert x\vert}{\theta}\right)^{\beta -1}e^{-\left(\frac{\vert x\vert}{\theta}\right)^{\beta}}
  \]
  in $(\Omega, \mathfrak{F}, \mathbf{P}_1)$, where
  \[
    \theta = \sqrt{\frac{1}{\Gamma \left(1+ \frac{2}{\beta}\right)}},
  \]
   and $\beta \in (0, 1]$, and Rademacher distribution (takes values -1 and 1 with the same probability) in $(\Omega, \mathfrak{F}, \mathbf{P}_2)$, and every random variable from a sequence $\{X_{2i}\}_{i=1}^{\infty}$ has Rademacher distribution in $(\Omega, \mathfrak{F}, \mathbf{P}_1)$, and absolutely continuous probability distribution with pdf $f_{X}(x)$ in $(\Omega, \mathfrak{F}, \mathbf{P}_2)$. Thus for all random variables of the sequence $\{X_i\}_{i=1}^{\infty}$ we have $\mathbf{E}_1 X_i = \mathbf{E}_2 X_i  =0$ and $\mathbf{E}_1 X_i^2 = \mathbf{E}_2 X_i^2  =1$.\par
  We define a sequence of random variables $Y_n$ in $(\Omega, \mathcal{H}, \mathbb{E})$ as follows: $Y_n := X_{n} + X_{n+1}$.\par
  Using remark \ref{from_2_3}, we have
  \[
    \begin{aligned}
      &\mathbb{E}[Y_n] = \mathbb{E}[X_n + X_{n+1}]\leqslant \mathbb{E}[X_n] + \mathbb{E}[X_{n+1}]=0,\\
      &\mathbb{E}[Y_n] = \mathbb{E}[X_n - (-X_{n+1})]\geqslant \mathbb{E}[X_n] - \mathbb{E}[-X_{n+1}] = 0,
    \end{aligned}
  \]
  which implies
  \[
    \mathbb{E}[Y_n] = \mathbb{E}[-Y_n]=0,
  \]
  for all $n$.\par
  Due to independence of $\{X_i\}_{i=1}^{\infty}$, for any fixed $n$ we have
  \begin{multline*}
    \mathbf{E}_j [Y_1 + \ldots + Y_n]^2 = \mathbf{E}_j [X_1 + 2X_2 \ldots + 2X_{n} + X_{n+1}]^2\\
    = \mathbf{E}_j [X_1^2] + \mathbf{E}_j [X_{n+1}^2] + 4\sum\limits_{i=2}^{n}\mathbf{E}_j [X_i^2] = 2 + 4(n-1),\quad j =1,2,
  \end{multline*}
  which implies
  \[
    \lim\limits_{n\to\infty} \frac{1}{n}\mathbb{E}\left[\sum\limits_{i=1}^{n}Y_i\right]^2 = 4.
  \]\par
  It is easy to see that $\{Y_i\}_{i=1}^{\infty}$ is a sequence of $m$--dependent random variables with $m=1$.
  \par Let $\varphi$ be a Borel measurable function on $\mathbb{R}^n$, such that $\varphi(Y_{1},\ldots, Y_{n}) \in \mathcal{H}$. It is easy to see that for any $k$ we have
  \[
    \max\limits_{i\in \{1,2\}}\mathbf{E}_i \varphi(Y_{1},\ldots, Y_{n}) =\max\limits_{i\in \{1,2\}}\mathbf{E}_i \varphi(Y_{1+k},\ldots, Y_{n+k}).
  \]
  Thus, the sequence $\{Y_i\}_{i=1}^{\infty}$ is strictly stationary.
  \par
  Every random variable of the sequence $\{Y_i\}_{i=1}^{\infty}$ has absolutely continuous distribution in probability spaces $(\Omega, \mathfrak{F}, \mathbf{P}_1)$ and $(\Omega, \mathfrak{F}, \mathbf{P}_2)$ with the following pdf
  \[
    f_{Y}(x) = \frac{1}{2}\Big(f_{X}(x-1) + f_{X}(x+1)\Big),
  \]
  from which it follows that for any $q>0$ and $\alpha < \beta$ we have
  \[
    \mathbb{E}[e^{q\vert Y_n\vert^{\alpha}}] \leqslant C,\quad n\in\mathbb{N}.
  \]
  Thus the theorem \ref{mdp_theorem} is applicable to the sequence $\{Y_i\}_{i=1}^{\infty}$ with the rate function
  \[
    I(y) = \frac{y^2}{8}.
  \]
\end{example*}

The rest of this paper consists of sections 2 and 3.
In section 2 we prove theorem \ref{mdp_theorem}, and in section 3 we formulate and prove auxiliary results.

\section{Proof of theorem \ref{mdp_theorem}}
\begin{proof}[\unskip\nopunct]
  In this section we prove the main result of the paper. The proof is devided into two steps: first we prove inequality (\ref{mdp_def_upper}), then we prove inequality (\ref{mdp_def_lower}) (see definition \ref{mdp_def}).\par
  \textbf{Step 1.} Let $F$ be any given closed set. If $F=\varnothing$ then the result is obvious. Suppose that $F\neq\varnothing$. Denote
    \[
      y_{-} := \sup \{y\in F:y<0\}\leqslant 0,\quad y_{+}:= \inf \{y\in F: y\geqslant 0\}\geqslant 0.
    \]
  It is easy to see that $F\subseteq (-\infty, y_{-}]\cup [y_{+},+\infty)$. Here we assume that $y_{-} = -\infty$, if $F\cap (-\infty,0] = \varnothing$ and $y_{+} = +\infty$, if $F\cap [0,+\infty) = \varnothing$. It is easy to see that if $F \neq \varnothing$, then at least one of the values $y_{-}$ and $y_{+}$ is finite. We have that
  \begin{align}\label{ln_V_upper}
    \ln \mathbb{V} (s_n \in F) & = \ln \mathbb{E} [\mathbf{I}(s_n\in F)]\nonumber\\
    & \leqslant \ln \Big(\mathbb{E} \big[\mathbf{I}(s_n \in (-\infty, y_{-}])\big] + \mathbb{E} \big[\mathbf{I}(s_n\in[y_{+},+\infty))\big]\Big)\nonumber\\
    & \leqslant \ln \Big(2\max\big(\mathbb{E} \big[\mathbf{I}(s_n \in (-\infty, y_{-}])\big], \mathbb{E} \big[\mathbf{I}(s_n\in[y_{+},+\infty))\big]\big) \Big)
  \end{align}\par
  Denote
  \[
    A_n := \left\{\omega : \max\limits_{1\leqslant i \leqslant n} \vert X_i \vert \leqslant x(n)\right\}.
  \]\par
  Now we find the upper bound for $\mathbb{E} \big[\mathbf{I}(s_n \in (-\infty, y_{-}])\big]$. Using Chebyshev inequality (see lemma \ref{geld_cheb}, (ii)) with the function $f(x) = x$, $x>0$, for any $\lambda >0$ we have that
  \begin{align*}
    \mathbb{E} [\mathbf{I}(s_n \in (-\infty, y_{-}])] &\leqslant \mathbb{E} \big[\mathbf{I}(s_n \in (-\infty, y_{-}])\mathbf{I}(A_n)\big] + \mathbb{E} [\mathbf{I}(\overline{A}_n)]\\
    &\leqslant \mathbb{E} \left[\mathbf{I}\left(-\lambda \frac{x(n)}{n}S_n\geqslant -\lambda \frac{x^2(n)}{n}y_{-}\right)\mathbf{I}(A_n)\right] +  \mathbb{E} [\mathbf{I}(\overline{A}_n)]\\
    &= \mathbb{E} \left[\mathbf{I}\left(e^{-\lambda \frac{x(n)}{n}S_n}\mathbf{I}(A_n)\geqslant e^{-\lambda \frac{x^2(n)}{n}y_{-}}\right)\mathbf{I}(A_n)\right] +  \mathbb{E} [\mathbf{I}(\overline{A}_n)]\\
    &\leqslant \mathbb{E} \left[\mathbf{I}\left(e^{-\lambda \frac{x(n)}{n}S_n}\mathbf{I}(A_n)\geqslant e^{-\lambda \frac{x^2(n)}{n}y_{-}}\right)\right] +  \mathbb{E} [\mathbf{I}(\overline{A}_n)]\\
    &\leqslant \frac{\mathbb{E} \left[e^{-\lambda \frac{x(n)}{n}S_n}\mathbf{I}(A_n)\right]}{e^{-\lambda \frac{x^2(n)}{n}y_{-}}} +  n\mathbb{E} \big[\mathbf{I}(\vert X_1\vert > x(n))\big].
  \end{align*}
  Using Chebyshev inequality (see lemma \ref{geld_cheb}, (ii)) with the function $f(x) = e^{qx^{\alpha}}$ and moment condition (\ref{exp_moment}), we have that
  \begin{equation}\label{ind_upper}
    n\mathbb{E} \big[\mathbf{I}(\vert X_1\vert > x(n))\big] \leqslant n\frac{\mathbb{E} [e^{q \vert X_1\vert^{\alpha}}]}{e^{qx(n)^{\alpha}}} \leqslant Mn e^{-q x^{\alpha}(n)}.
  \end{equation}
  From lemma \ref{eq_lemma} it follows that
  \begin{equation}\label{eq_lemma_alt}
    \mathbb{E} \left[e^{\lambda \frac{x(n)}{n}S_n}\mathbf{I}(A_n)\right] = e^{\frac{x^2(n)}{n}\left(\frac{\lambda^2}{2}\sigma^2 + o(1)\right)},
  \end{equation}
  as $n\rightarrow\infty$.
  From (\ref{ind_upper}) and (\ref{eq_lemma_alt}) it follows that
  \[
    \mathbb{E} \big[\mathbf{I}(s_n \in (-\infty, y_{-}])\big]\leqslant e^{\frac{x^2(n)}{n}\left(\frac{\lambda^2}{2}\sigma^2 + \lambda y_{-} + o(1)\right)} + Mne^{-qx(n)^{\alpha}}.
  \]
  By choosing $\lambda = - \frac{y_{-}}{\sigma^2}$, we get
  \begin{equation}\label{y_minus}
    \mathbb{E} \big[\mathbf{I}(s_n \in (-\infty, y_{-}])\big]\leqslant e^{-\frac{x^2(n)}{n}\left(\frac{y^2_{-}}{2\sigma^2} + o(1)\right)} + Mne^{-qx(n)^{\alpha}}.
  \end{equation}
  By using the similar method we can get
  \begin{equation}\label{y_plus}
    \mathbb{E} \big[\mathbf{I}(s_n \in [y_{+},+\infty))\big]\leqslant e^{-\frac{x^2(n)}{n}\left(\frac{y^2_{+}}{2\sigma^2} + o(1)\right)} + Mne^{-qx(n)^{\alpha}}.
  \end{equation}
  The maximum of right-hand sides of inequalities (\ref{y_minus}) and (\ref{y_plus}) is determined by the minimum of $y^2_{-}$ and $y^2_{+}$. Suppose that $y^2_{-}\leqslant y^2_{+}$. We have

  \begingroup
  \allowdisplaybreaks
  \begin{multline*}
  \begin{aligned}
    &\limsup\limits_{n\to\infty}\frac{n}{x^2(n)}\ln \left(e^{-\frac{x^2(n)}{n}\left(\frac{y^2_{-}}{2\sigma^2} + o(1)\right)} + Mne^{-qx(n)^{\alpha}}\right)\\
    &=\limsup\limits_{n\to\infty}\frac{n}{x^2(n)}\ln \left(\left(e^{-\frac{x^2(n)}{n}\left(\frac{y^2_{-}}{2\sigma^2} + o(1)\right)}\right)\left(1 + e^{\ln(Mn)-qx(n)^{\alpha} + \frac{x^2(n)}{n}\left(\frac{y^2_{-}}{2\sigma^2} + o(1)\right)}\right)\right)\\
    &=\limsup\limits_{n\to\infty} \left(-\left(\frac{y^2_{-}}{2\sigma^2} + o(1)\right) + \frac{n}{x^2(n)}\ln \left( 1 + e^{\ln(Mn)-qx(n)^{\alpha} + \frac{x^2(n)}{n}\left(\frac{y^2_{-}}{2\sigma^2} + o(1)\right)}\right)\right)\\
    &=\limsup\limits_{n\to\infty} \left(-\left(\frac{y^2_{-}}{2\sigma^2} + o(1)\right)\right.\\
    &\hphantom{+}\hphantom{+}\hphantom{+}\hphantom{+}\hphantom{+}\hphantom{+}\hphantom{+}\hphantom{+}\hphantom{+}\hphantom{+}\hphantom{+}\hphantom{+}\left. + \frac{n}{x^2(n)}\ln \left( 1 + e^{-qx(n)^{\alpha}\left(1 -\frac{\ln(Mn)}{qx(n)^{\alpha}} - \frac{x^{2-\alpha}(n)}{qn}\left(\frac{y^2_{-}}{2\sigma^2} + o(1)\right)\right)}\right)\right)\\
  \end{aligned}\\
  =-\frac{y^2_{-}}{2\sigma^2}.
  \end{multline*}
  \endgroup
  Similarly by assuming $y^2_{+}\leqslant y^2_{-}$ we get
  \[
    \limsup\limits_{n\to\infty}\frac{n}{x^2(n)}\ln \left(e^{-\frac{x^2(n)}{n}\left(\frac{y^2_{+}}{2\sigma^2} + o(1)\right)} + Mne^{-qx(n)^{\alpha}}\right) = -\frac{y^2_{+}}{2\sigma^2}.
  \]
  Thus we get
  \[
    \limsup\limits_{n\to\infty} \frac{n}{x^2(n)}\ln \big(\mathbb{V}(s_n\in F)\big)\leqslant -\min \left( \frac{y^2_{-}}{2\sigma^2},\frac{y^2_{+}}{2\sigma^2} \right).
  \]
  It remains to notice, that
  \[
    \inf\limits_{y\in F} I(y) = \min(I(y_{-}),I(y_{+})) = \min \left( \frac{y^2_{-}}{2\sigma^2},\frac{y^2_{+}}{2\sigma^2} \right).
  \]
  \par
  \textbf{Step 2.} Let $G$ be any given open set. If $G = \varnothing$, then the result is obvious. Suppose that $G\neq\varnothing$. It is easy to see, that for any $l\geqslant 0$ the set
  \[
    K_l := \{y:I(y) \leqslant l\}
  \]
  is compact. Since $G\neq\varnothing$, there exists $l_G >0$ such that $G\cap K_{l_G} \neq \varnothing$.
  \par
  Since $G$ is an open set and $\inf\limits_{y\in G\cap K_{l_G}}I(y) = \inf\limits_{y\in G}I(y)$, for any $\varepsilon > 0$ there exists $y\in G \cap K_{l_{G}}$ such that
  \begin{equation}\label{I_minus_vareps}
    \inf\limits_{x\in G}I(x)\geqslant I(y) -\varepsilon.
  \end{equation}
  For any $\delta > 0$ denote
  \[
    y^{(\delta)}:= (y-\delta,y+\delta).
  \]
  Since $G$ is an open set, for small enough $\delta$ we have
  \begin{equation}\label{V_equ}
    \mathbb{V}(s_n \in G)\geqslant \mathbb{E} \left[\mathbf{I}(s_n\in G)\mathbf{I}(A_n)\right] \geqslant \mathbb{E} \left[\mathbf{I}(s_n\in y^{(\delta)})\mathbf{I}(A_n)\right].
  \end{equation}
  For $\lambda \in \mathbb{R}$ and $n \in \mathbb{N}$ we denote
  \[
    A(\lambda,n):= \ln \mathbb{E} \left[e^{\lambda \frac{x(n)}{n}S_n}\mathbf{I}(A_n)\right].
  \]
  From definition of $y^{(\delta)}$ it follows that
  \begin{multline}\label{A_ln}
    \mathbf{I}(s_n\in y^{(\delta)})\mathbf{I}(A_n)e^{\lambda \frac{x(n)}{n}S_n} = \mathbf{I}(s_n\in y^{(\delta)})\mathbf{I}(A_n)e^{\lambda \frac{x^2(n)}{n}s_n}\\
    \leqslant \mathbf{I}(s_n\in y^{(\delta)})\mathbf{I}(A_n)e^{\frac{x^2(n)}{n}(\lambda y + \vert\lambda\vert\delta)}.
  \end{multline}
  Using remark \ref{from_2_3} and inequality (\ref{A_ln}), for any $\lambda \in \mathbb{R}$ and small enough $\delta > 0$ we have that
  \begingroup
  \allowdisplaybreaks
  \begin{align}\label{lnV_equ}
    &\ln\mathbb{E} \left[\mathbf{I}(s_n\in y^{(\delta)})\mathbf{I}(A_n)\right] \nonumber\\
    &\geqslant\ln\mathbb{E} \left[\mathbf{I}(s_n\in y^{(\delta)})\mathbf{I}(A_n)e^{\lambda \frac{x(n)}{n}S_n - A(\lambda,n)}e^{-\lambda \frac{x^2(n)}{n}y + A(\lambda,n)}e^{-\vert\lambda\vert \frac{x^2(n)}{n}\delta}\right]\nonumber\\
    &= -\lambda \frac{x^2(n)}{n}y + A(\lambda,n) -\vert\lambda\vert \frac{x^2(n)}{n}\delta\nonumber\\
    &\hphantom{+}\hphantom{+}\hphantom{+}\hphantom{+}\hphantom{+}\hphantom{+}\hphantom{+}\hphantom{+}\hphantom{+}+ \ln\mathbb{E} \left[\big(1-\mathbf{I}(s_n\notin y^{(\delta)})\big)\mathbf{I}(A_n)e^{\lambda \frac{x(n)}{n}S_n - A(\lambda,n)}\right]\nonumber\\
    &\geqslant -\lambda \frac{x^2(n)}{n}y + A(\lambda,n) -\vert\lambda\vert \frac{x^2(n)}{n}\delta\nonumber\\
    &\hphantom{+}\hphantom{+}\hphantom{+}\hphantom{+}\hphantom{+}\hphantom{+}\hphantom{+}\hphantom{+}\hphantom{+}+ \ln\left(1-\mathbb{E} \left[\mathbf{I}(s_n\notin y^{(\delta)})\mathbf{I}(A_n)e^{\lambda \frac{x(n)}{n}S_n - A(\lambda,n)}\right]\right).
  \end{align}
  \endgroup
  For any $r>0$ we have
  \begin{align*}
    &\mathbb{E} \left[\mathbf{I}(s_n \notin y^{(\delta)})\mathbf{I}(A_n)e^{\lambda \frac{x(n)}{n}S_n - A(\lambda, n)}\right]\\
    & \leqslant \mathbb{E} \left[\mathbf{I}(s_n \geqslant y+\delta)\mathbf{I}(A_n)e^{\lambda \frac{x(n)}{n}S_n - A(\lambda, n)}\right] + \mathbb{E} \left[\mathbf{I}(s_n \leqslant y-\delta)\mathbf{I}(A_n)e^{\lambda \frac{x(n)}{n}S_n - A(\lambda, n)}\right]\\
    & \leqslant \mathbb{E} \left[\frac{e^{\frac{rx^2(n)}{n\sigma^2}(s_n - (y+\delta))}\mathbf{I}(A_n)e^{\lambda \frac{x(n)}{n}S_n}}{e^{A(\lambda,n)}}\right] + \mathbb{E} \left[\frac{e^{\frac{rx^2(n)}{n\sigma^2}((y-\delta)-s_n)}\mathbf{I}(A_n)e^{\lambda \frac{x(n)}{n}S_n}}{e^{A(\lambda,n)}}\right]\\
    &=\frac{\mathbb{E} \left[e^{\frac{rx(n)}{n\sigma^2}S_n}\mathbf{I}(A_n)e^{\lambda \frac{x(n)}{n}S_n}\right]}{e^{\frac{rx^2(n)}{n\sigma^2}(y+\delta)}e^{A(\lambda,n)}} + \frac{\mathbb{E} \left[e^{-\frac{rx(n)}{n\sigma^2}S_n}\mathbf{I}(A_n)e^{\lambda \frac{x(n)}{n}S_n}\right]}{e^{\frac{rx^2(n)}{n\sigma^2}(\delta-y)}e^{A(\lambda,n)}}\\
    &=\frac{\mathbb{E} \left[e^{\left(\frac{r}{\sigma^2} + \lambda\right)\frac{x(n)}{n}S_n}\mathbf{I}(A_n)\right]}{e^{\frac{rx^2(n)}{n\sigma^2}(y+\delta)}e^{A(\lambda,n)}} + \frac{\mathbb{E} \left[e^{\left(\lambda -\frac{r}{\sigma^2} \right)\frac{x(n)}{n}S_n}\mathbf{I}(A_n)\right]} {e^{\frac{rx^2(n)}{n\sigma^2}(\delta-y)}e^{A(\lambda,n)}}\\
    &= I_1 + I_2.
  \end{align*}
  Now we find an upper bound for $I_1$. From (\ref{eq_lemma_alt}) it follows that
  \[
    I_1 \leqslant \frac{e^{\frac{x^2(n)}{n}\left(\frac{(r/\sigma^2 + \lambda)^2}{2}\sigma^2 + o(1)\right)}}{e^{\frac{x^2(n)}{n}\left(\frac{r}{\sigma^2}(y+\delta) +\frac{\lambda^2}{2}\sigma^2\right)}}
    = \exp \left\{\frac{x^2(n)}{n} \left(\frac{r^2}{2\sigma^2} - \frac{r\delta}{\sigma^2} + r \left(\lambda - \frac{y}{\sigma^2}\right) + o(1)\right)\right\},
  \]
  as $n\to\infty$. If we choose $r=\delta$, then
  \[
    I_1 \leqslant \exp \left\{\frac{x^2(n)}{n} \left(-\frac{\delta^2}{2\sigma^2}+ \delta \left(\lambda - \frac{y}{\sigma^2}\right) + o(1)\right)\right\},
  \]
  as $n\to\infty$. Similarly we get
  \[
    I_2 \leqslant \exp \left\{\frac{x^2(n)}{n} \left(-\frac{\delta^2}{2\sigma^2}+ \delta \left(\frac{y}{\sigma^2}-\lambda\right) + o(1)\right)\right\},
  \]
  as $n\to\infty$. Thus we have
  \begin{multline}
    \mathbb{E} \left[\mathbf{I}(s_n \notin y^{(\delta)})\mathbf{I}(A_n)e^{\lambda \frac{x(n)}{n}S_n - A(\lambda, n)}\right]\\
    \leqslant 2\exp \left\{\frac{x^2(n)}{n} \left(-\frac{\delta^2}{2\sigma^2}+ \delta \left\vert\frac{y}{\sigma^2}-\lambda\right\vert + o(1)\right)\right\},
  \end{multline}
  as $n\to\infty$.\par
  Let $\lambda = \frac{y}{\sigma^2}$. Thus using (\ref{eq_lemma_alt}) we get
  \begin{equation}\label{A_ln2}
    \frac{n}{x^2(n)}A \left(\frac{y}{\sigma^2},n\right) = \frac{n}{x^2(n)}\ln e^{\frac{x^2(n)}{n}\left(\frac{y^2}{2\sigma^2}+o(1)\right)} = \frac{y^2}{2\sigma^2} + o(1),
  \end{equation}
  as $n\to\infty$. We also get
  \begin{equation}\label{s_n_not_in_upper}
    \mathbb{E} \left[\mathbf{I}(s_n \notin y^{(\delta)})\mathbf{I}(A_n)e^{\lambda \frac{x(n)}{n}S_n - A(\lambda, n)}\right] \leqslant 2\exp \left\{\frac{x^2(n)}{n} \left(-\frac{\delta^2}{2\sigma^2} + o(1)\right)\right\},
  \end{equation}
  as $n\to\infty$. Using (\ref{lnV_equ}), (\ref{A_ln2}) and (\ref{s_n_not_in_upper}), we get that
  \begingroup
  \allowdisplaybreaks
  \begin{align*}
    \liminf\limits_{n\to\infty}\frac{n}{x^2(n)} \ln\mathbb{E} \left[\mathbf{I}(s_n\in y^{(\delta)})\mathbf{I}(A_n)\right]&
    \geqslant \liminf\limits_{n\to\infty}\frac{n}{x^2(n)} \left(-\frac{x^2(n)}{n\sigma^2}y^2 + A \left(\frac{y}{\sigma^2},n\right)\right.\\
    &\left. -\frac{\vert y\vert}{\sigma^2} \frac{x^2(n)}{n}\delta + \ln\left(1-2e^{\frac{x^2(n)}{n}\left(-\frac{\delta^2}{2\sigma^2} + o(1)\right)}\right)\right)\\
    &=-\frac{y^2}{2\sigma^2} - \frac{\vert y \vert}{\sigma^2}\delta.
  \end{align*}
  \endgroup

  That is
  \[
    \liminf\limits_{n\to\infty}\frac{n}{x^2(n)}\ln \mathbb{V}(s_n\in G) \geqslant -\frac{y^2}{2\sigma^2} - \frac{\vert y \vert}{\sigma^2}\delta.
  \]
  Since the left-hand side of the above inequality does not depend on $\delta$, by letting $\delta\to 0$, we get
  \[
    \liminf\limits_{n\to\infty}\frac{n}{x^2(n)} \ln\mathbb{V}(s_n\in G) \geqslant -\frac{y^2}{2\sigma^2}.
  \]
  By using (\ref{I_minus_vareps}), we get
  \[
    \liminf\limits_{n\to\infty}\frac{n}{x^2(n)} \ln\mathbb{V}(s_n\in G) \geqslant -\frac{y^2}{2\sigma^2} = -I(y)\geqslant - \inf\limits_{x\in G}I(x) - \varepsilon.
  \]
  It remains to notice that the left-hand side of the above inequality does not depend on $\varepsilon$. By letting $\varepsilon\to0$ we complete the proof.
\end{proof}

\section{Auxiliary results}
In this section we formulate auxiliary results (lemmas 1-3). Lemmas \ref{geld_cheb} and \ref{liu_lemma} are proven in works of other authors. For lemma \ref{eq_lemma} a proof will be given.
\begin{lemma}\label{geld_cheb} Let $X, Y \in \mathcal{H}$. Then the following statements are true
  \begin{enumerate}
    \item[(i)] H\"{o}lder inequality: Let $p,q > 1$ and $\frac{1}{p} + \frac{1}{q}=1$. Let $\vert XY\vert \in \mathcal{H}$, $\vert X\vert^p \in \mathcal{H}$ and $\vert Y\vert^q\in\mathcal{H}$. Then
    \[
      \mathbb{E}[\vert XY\vert]\leqslant (\mathbb{E}[\vert X\vert^p])^{\frac{1}{p}}\cdot (\mathbb{E}[\vert Y\vert^q])^{\frac{1}{q}}.
    \]
    If $p=q=2$, then we have Cauchy--Bunyakovsky--Schwarz inequality.
    \item[(ii)] Chebyshev inequality: Let $f(x)$ be a nondecreasing nonnegative function on $\mathbb{R}$ and $f(X)\in \mathcal{H}$. Then for any $x>0$
    \[
      \mathbb{V}(X\geqslant x)\leqslant \frac{\mathbb{E}[f(X)]}{f(x)}.
    \]
  \end{enumerate}
\end{lemma}
The proof of lemma \ref{geld_cheb} can be found in \cite{chen}.
\begin{lemma}\label{liu_lemma}
Let $X_1,\ldots,X_n$ be random variables on $(\Omega, \mathcal{H},\mathbb{E})$ and $\zeta_i \in [0,1]$, $1\leqslant i \leqslant n$ such that $\sum\limits_{i=1}^{n}\zeta_i = 1$. Then
\[
  \ln\mathbb{E}\left[e^{\sum\limits_{i=1}^{n}\zeta_iX_i }\right] \leqslant \sum\limits_{i=1}^{n}\zeta_i\ln\mathbb{E}\left[ e^{X_i}\right].
\]
\end{lemma}
The proof of lemma \ref{liu_lemma} can be found in \cite{liu}.

\begin{lemma}\label{eq_lemma}
For any $\lambda \in \mathbb{R}$ the following equality holds
\[
  \lim\limits_{n\to\infty} \frac{n}{x^2(n)}\ln\mathbb{E} \left[ e^{\lambda \frac{x(n)}{n}S_n}\mathbf{I}(A_n) \right] = \frac{\sigma^2\lambda^2}{2}.
\]
\end{lemma}
\begin{proof}
  Fix $K \geqslant m + 1,\; n \geqslant K+m$. Denote $l := \left[ \frac{n}{K+m}\right]$. Define sequences
  \[
    \xi_t := \sum\limits_{i=1}^{K}X_{t(K+m) +i},\quad \eta_t := \sum\limits_{i=1}^{m}X_{t(K+m) +K + i},\quad t = 0,1,\ldots, l-1.
  \]
  Since the sequence $\{X_i\}_{i=1}^{\infty}$ is $m$-dependent, we have that $\xi_t$ is independent from $\xi_1,\ldots,\xi_{t-1}$, $1\leqslant t\leqslant l-1$, and $\eta_t$ is independent from $\eta_1,\ldots,\eta_{t-1}$, $1\leqslant t\leqslant l-1$.\par
  The sequence $S_n$ can be represented as follows
  \[
    S_n = \sum\limits_{t=0}^{l-1}\xi_t + \sum\limits_{t=0}^{l-1}\eta_t + \sum\limits_{i=l(K+m)+1}^{n}X_i = I_1 + I_2 + I_3.
  \]\par
  The rest of the proof of lemma \ref{eq_lemma} relies on the following three propositions, which we formulate and prove below.\par\noindent
  \textbf{Proposition 1.} For any fixed $y\in\mathbb{R}$, $K\geqslant m+1$ the following equality holds
  \[
    \lim\limits_{n\to\infty} \frac{n}{x^2(n)}\ln\mathbb{E} \left[e^{y \frac{x(n)}{n}I_1}\mathbf{I}(A_n)\right] = \frac{1}{K+m} \frac{y^2\sigma^2_0}{2},
  \]
  where $\sigma^2_0 = \mathbb{E} \left[\xi^2_0\right]$.\par\noindent
  \textit{Proof of proposition 1}. Define
  \[
    \Delta_t = \left\{i\in\mathbb{N} : 1+t(K+m)\leqslant i\leqslant K+t(K+m) \right\},
  \]
  \[
    \widetilde{\Delta}_n = \left\{i\in\mathbb{N} : 1 \leqslant i \leqslant n \right\}\setminus \left(\bigcup\limits_{t=1}^{l-1}\Delta_t\right),
  \]
  \[
    B_t = \left\{\omega : \max\limits_{i\in\Delta_t}\vert X_i\vert \leqslant x(n)\right\},
  \]
  \[
    C_n = \left\{\omega : \max\limits_{i\in\widetilde{\Delta}_n}\vert X_i\vert \leqslant x(n)\right\}.
  \]
  Using the fact, that $\xi_t$ is independent from $\xi_1,\ldots,\xi_{t-1}$, $1\leqslant t\leqslant l-1$, and the fact, that $\{X_i\}_{i=1}^{\infty}$ is strictly stationary, we get
  \begingroup
  \allowdisplaybreaks
  \begin{multline}\label{eq1}
     \mathbb{E} \left[e^{y \frac{x(n)}{n}I_1}\mathbf{I}(A_n)\right]= \mathbb{E} \left[e^{y \frac{x(n)}{n}I_1}\mathbf{I}(C_n)\prod\limits_{t=1}^{l-1}\mathbf{I}(B_t)\right]\\
    \leqslant \mathbb{E} \left[e^{y \frac{x(n)}{n}I_1}\prod\limits_{t=1}^{l-1}\mathbf{I}(B_t)\right]=\mathbb{E} \left[\prod\limits_{t=1}^{l-1}e^{y \frac{x(n)}{n}\xi_t}\mathbf{I}(B_t)\right]\\
    =\prod\limits_{t=1}^{l-1}\mathbb{E}\left[e^{y \frac{x(n)}{n}\xi_t}\mathbf{I}(B_t)\right]= \left(\mathbb{E}\left[e^{y \frac{x(n)}{n}\xi_0}\mathbf{I}(B_0)\right]\right)^{l}.
  \end{multline}
  \endgroup
  Using also the remark \ref{from_2_3}, we get
  \begingroup
  \allowdisplaybreaks
  \begin{align}\label{eq2}
    \mathbb{E} \left[e^{y \frac{x(n)}{n}I_1}\mathbf{I}(A_n)\right] & = \mathbb{E} \left[e^{y \frac{x(n)}{n}I_1}\mathbf{I}(C_n)\prod\limits_{t=1}^{l-1}\mathbf{I}(B_t)\right]\nonumber\\
    &\geqslant \mathbb{E}\left[e^{y \frac{x(n)}{n}I_1}\prod\limits_{t=1}^{l-1}\mathbf{I}(B_t)\right] - \mathbb{E}\left[e^{y \frac{x(n)}{n}I_1}\mathbf{I}(\overline{C}_n)\prod\limits_{t=1}^{l-1}\mathbf{I}(B_t)\right]\nonumber\\
    &= \left(\mathbb{E}\left[e^{y \frac{x(n)}{n}\xi_0}\mathbf{I}(B_0)\right]\right)^{l} - \mathbb{E}\left[\mathbf{I}(\overline{C}_n)\prod\limits_{t=1}^{l-1}e^{y \frac{x(n)}{n}\xi_t}\mathbf{I}(B_t)\right].
  \end{align}
  \endgroup

  Now we show that for any $y\in\mathbb{R}$ the following holds
  \begin{equation}\label{eq3}
    \mathbb{E} \left[e^{y \frac{x(n)}{n}\xi_0}\mathbf{I}(B_0)\right] = \exp\left\{\frac{x^2(n)}{n^2} \left(\frac{\sigma^2_0y^2}{2} + o(1)\right)\right\},
  \end{equation}
  as $n\to\infty$.\par
  Let us find an upper bound for
 \begin{equation*}
   \mathbb{E} \left[e^{ y  \frac{x(n)}{n}\xi_0}\mathbf{I}( B _0)\right].
 \end{equation*}
 Using Taylor expansion of $e^x$, we get
 \begingroup
 \allowdisplaybreaks
 \begin{multline*}
   \mathbb{E} \left[e^{y \frac{x(n)}{n}\xi_0}\mathbf{I}( B _0)\right]= \mathbb{E} \left[\left(1 +  y  \frac{x(n)}{n}\xi_0 +  y ^2 \frac{x^2(n)}{2n^2}\xi^2_0 + \sum\limits_{r=3}^{+\infty} y ^r \frac{x^r(n)}{r!n^r}\xi^r_0\right)\mathbf{I}( B _0)\right] \\
   = \mathbb{E} \left[\mathbf{I}(B_0) +  y  \frac{x(n)}{n}\xi_0 - y  \frac{x(n)}{n}\xi_0\mathbf{I}(\overline{ B }_0)  +  y ^2 \frac{x^2(n)}{2n^2}\xi^2_0\mathbf{I}( B _0) + \sum\limits_{r=3}^{+\infty} y ^r \frac{x^r(n)}{r!n^r}\xi^r_0\mathbf{I}( B _0)\right]\\
   \leqslant 1 + \mathbb{E}\left[\vert  y \vert \frac{x(n)}{n}\vert \xi_0\vert \mathbf{I}(\overline{ B }_0)\right] +\mathbb{E} \left[ y ^2 \frac{x^2(n)}{2n^2}\xi^2_0\mathbf{I}( B _0)\right]\\
   + \mathbb{E} \left[\sum\limits_{r=3}^{+\infty} \frac{\vert  y  \xi_0\vert^rx^r(n)}{r!n^r}\mathbf{I}( B _0)\right]\\
   = 1 + \mathbb{E}_1 + \mathbb{E} \left[ y ^2 \frac{x^2(n)}{2n^2}\xi^2_0\mathbf{I}( B _0)\right] + \mathbb{E}_2.
 \end{multline*}
 \endgroup
 Now we find an upper bound for $\mathbb{E}_1$. First, by using Cauchy--Bunyakovsky--Schwarz inequality (see lemma \ref{geld_cheb}, (i)), we get
  \begin{align*}
    \mathbb{E}\left[\vert  y  \vert \frac{x(n)}{n}\vert \xi_0\vert \mathbf{I}(\overline{ B }_0)\right] &\leqslant \left(\mathbb{E}\left[  y ^2 \frac{x^2(n)}{n^2} \xi_0^2\right]\right)^{\frac{1}{2}} \Big(\mathbb{V}(\overline{ B }_0)\Big)^{\frac{1}{2}}\\
    &= \vert  y  \vert \frac{x(n)}{n}(\sigma^2_0)^{\frac{1}{2}}\Big(\mathbb{V}(\overline{ B }_0)\Big)^{\frac{1}{2}}\\
    &\leqslant \vert  y \vert \frac{x(n)}{n}(\sigma^2_0)^{\frac{1}{2}}\Big(K\mathbb{V}(\vert X_1\vert > x(n))\Big)^{\frac{1}{2}}.
  \end{align*}
  Then, by using Chebyshev inequality (see lemma \ref{geld_cheb}, (ii)) with the function $f(x) = e^{q x^{\alpha}}$, property (\ref{homogen}) and moment condition (\ref{exp_moment}), we get
  \begin{multline}
    \vert  y \vert \frac{x(n)}{n}(\sigma^2_0)^{\frac{1}{2}}\Big(K\mathbb{V}(\vert X_1\vert > x(n))\Big)^{\frac{1}{2}} \leqslant \vert  y \vert \frac{x(n)}{n}(\sigma^2_0)^{\frac{1}{2}} \frac{K^{\frac{1}{2}}\left(\mathbb{E} \left [e^{q \vert X_1\vert^{\alpha}} \right]\right)^{\frac{1}{2}} }{e^{\frac{q}{2} x^{\alpha}(n)}} \\
    \leqslant \vert  y \vert \frac{x(n)K^{\frac{1}{2}}M^{\frac{1}{2}}}{n e^{\frac{q}{2}x^{\alpha}(n)}}\left(\sigma^2_0 \right)^{\frac{1}{2}}.
  \end{multline}
  Thus we get the following bound
  \begin{equation}\label{E1_upper}
    \mathbb{E}_1 \leqslant \vert  y \vert \frac{x(n)K^{\frac{1}{2}}M^{\frac{1}{2}}}{n e^{\frac{q}{2}x^{\alpha}(n)}}\left(\sigma^2_0 \right)^{\frac{1}{2}}.
  \end{equation}
 Now we find an upper bound for $\mathbb{E}_2$.
 \begingroup
 \allowdisplaybreaks
 \begin{multline*}
   \sum\limits_{r=3}^{+\infty} \frac{\vert  y  \xi_0\vert^rx^r(n)}{r!n^r}\mathbf{I}( B _0) \leqslant \frac{\vert  y  \xi_0\vert^3x^3(n)}{3!n^3}\exp\left\{\frac{\vert  y  \xi_0\vert x(n)}{n}\right\}\mathbf{I}( B _0)\\
   \leqslant \frac{\vert  y \vert^3(\vert X_1\vert + \ldots +
   \vert X_K\vert)^3x^3(n)}{3!n^3}\exp\left\{\frac{x(n)\vert  y \vert(\vert X_1\vert + \ldots +\vert X_K\vert)}{n}\right\}\mathbf{I}( B _0)\\
   \leqslant\frac{\vert y \vert^3K^3\max\limits_{1\leqslant r\leqslant K}\vert X_r\vert^3 x^3(n)}{3!n^3}\exp \left\{ \frac{x(n)\vert y \vert K\max\limits_{1\leqslant r\leqslant K}\vert X_r\vert}{n}\right\}\mathbf{I}( B _0)\\
   \leqslant \sum\limits_{r=1}^{K}\frac{\vert y \vert^3K^3\vert X_r\vert^3 x^3(n)}{3!n^3}\exp \left\{ \frac{x(n)\vert y \vert K\vert X_r\vert}{n}\right\}\mathbf{I}( B _0).
 \end{multline*}
 \endgroup
 There exists $C>0$ such that for all $u>C$ the following inequality holds
 \begin{equation}\label{u3}
   \vert u \vert^3 \leqslant e^{\frac{q}{2}\vert u \vert^{\alpha}}.
 \end{equation}
 Using this fact, we proceed with the bound.
 \begingroup
 \allowdisplaybreaks
 \begin{multline*}
   \sum\limits_{r=3}^{+\infty} \frac{\vert  y  \xi_0\vert^rx^r(n)}{r!n^r}\mathbf{I}( B _0)\\
   \leqslant \sum\limits_{r=1}^{K}\frac{\vert y \vert^3K^3 C^3x^3(n)}{3!n^3}\exp \left\{\frac{q}{2}\vert X_r\vert^{\alpha} \right\}\exp \left\{ \frac{x(n)\vert y \vert K\vert X_r\vert}{n}\right\}\mathbf{I}( B _0)\\
   = \sum\limits_{r=1}^{K}\frac{\vert y \vert^3K^3 C^3x^3(n)}{3!n^3} \exp \left\{\vert X_r\vert^\alpha \left(\frac{q}{2} + \frac{x^{2-\alpha}(n)\vert  y \vert K }{n}\right) \right\}.
 \end{multline*}
 \endgroup
 From condition (\ref{seq}) it follows that for large enough $n$ the following inequality holds
 \[
   \frac{q}{2} + \frac{x^{2-\alpha}(n)\vert  y \vert K }{n} \leqslant q.
 \]
 By using this fact, moment condition (\ref{exp_moment}) and the fact that $\{X_i\}_{i=1}^{\infty}$ is strictly stationary, we get the following bound
 \begin{equation}\label{E2_upper}
 \mathbb{E}_2 \leqslant \frac{\vert y \vert^3K^4 C^3x^3(n)}{3!n^3}M.
 \end{equation}
 By using bounds (\ref{E1_upper}) and (\ref{E2_upper}) we get the following bound
 \begingroup
 \allowdisplaybreaks
 \begin{multline}\label{XI_upper}
   \mathbb{E} \left[e^{ y  \frac{x(n)}{n}\xi_0}\mathbf{I}( B _0)\right] \leqslant 1 +  \vert  y \vert \frac{x(n)K^{\frac{1}{2}}M^{\frac{1}{2}}}{n e^{\frac{q}{2}x^{\alpha}(n)}}\left(\sigma^2_0 \right)^{\frac{1}{2}}\\
   + \mathbb{E} \left[ y ^2 \frac{x^2(n)}{2n^2}\xi^2_0\mathbf{I}( B _0)\right] + \frac{\vert y \vert^3K^4 C^3x^3(n)}{3!n^3}M\\
   \leqslant 1 +  y ^2\frac{x^2(n)}{2n^2}\sigma^2_0 + o \left(\frac{x^2(n)}{n^2}\right),
 \end{multline}
 \endgroup
 as $n\to\infty$.\par
 Now we find a lower bound for
 \[
   \mathbb{E} \left[e^{ y  \frac{x(n)}{n}\xi_0}\mathbf{I}( B _0)\right].
 \]
 By using (\ref{E2_upper}) and remark \ref{from_2_3}, we get the following bound
 \begingroup
 \allowdisplaybreaks
 \begin{align*}
   &\mathbb{E} \left[e^{ y  \frac{x(n)}{n}\xi_0}\mathbf{I}( B _0)\right]\\ &\geqslant\mathbb{E} \left[\left(1 +  y  \frac{x(n)}{n}\xi_0 +  y ^2 \frac{x^2(n)}{2n^2}\xi_0^2 - \sum\limits_{r=3}^{+\infty} \frac{\vert  y  \xi_0\vert^r x^r(n)}{r!n^r}\right)\mathbf{I}( B _0)\right]\\
   &  =\mathbb{E} \left[ y ^2 \frac{x^2(n)}{2n^2}\xi_0^2\mathbf{I}( B _0) -\left(-1 +  y  \frac{x(n)}{n}(-\xi_0) + \sum\limits_{r=3}^{+\infty} \frac{\vert  y  \xi_0\vert^r x^r(n)}{r!n^r}\right)\mathbf{I}( B _0)\right]\\
   &  =\mathbb{E} \left[ y ^2 \frac{x^2(n)}{2n^2}\xi^2_0\mathbf{I}( B _0)\right.\\
   &\left.  -\left( y  \frac{x(n)}{n}(-\xi_0)\mathbf{I}( B _0)-1 +\mathbf{I}(\overline{ B }_0)+ \sum\limits_{r=3}^{+\infty} \frac{\vert  y  \xi_0\vert^r x^r(n)}{r!n^r}\mathbf{I}( B _0)\right)\right]\\
   &  \geqslant\mathbb{E} \left[ y ^2 \frac{x^2(n)}{2n^2}\xi_0^2\mathbf{I}( B _0) \right]\\
   &  -\mathbb{E}\left[\left( y  \frac{x(n)}{n}(-\xi_0)\mathbf{I}( B _0) +\mathbf{I}(\overline{ B }_0)\right)+\left( \sum\limits_{r=3}^{+\infty} \frac{\vert  y  \xi_0\vert^r x^r(n)}{r!n^r}\mathbf{I}( B _0)+(-1)\right)\right]\\
   &    \geqslant\mathbb{E} \left[ y ^2 \frac{x^2(n)}{2n^2}\xi^2_0\mathbf{I}( B _0)\right]\\
   &  -\mathbb{E}\left[ y  \frac{x(n)}{n}(-\xi_0)\mathbf{I}( B _0) + \mathbf{I}(\overline{ B }_0)\right] - \mathbb{E}\left[\sum\limits_{r=3}^{+\infty} \frac{\vert  y  \xi_0\vert^r x^r(n)}{r!n^r}\mathbf{I}( B _0)\right] + 1\\
   &    \geqslant \mathbb{E}_3 - \mathbb{E}_4 - \frac{\vert y \vert^3K^4 C^3x^3(n)}{3!n^3}M + 1.\\
 \end{align*}
 \endgroup
 Now get a lower bound for $\mathbb{E}_3$. By using remark \ref{from_2_3}, moment condition (\ref{exp_moment}), Cauchy--Bunyakovsky--Schwarz inequality and Chebyshev inequality with the function $f(x) = e^{q x^\alpha}$ (see lemma \ref{geld_cheb}, (i) and (ii) respectively) we get the following bound
 \begingroup
 \allowdisplaybreaks
 \begin{align}\label{E3_lower}
   \mathbb{E} \left[ y ^2 \frac{x^2(n)}{2n^2}\xi^2_0\mathbf{I}( B _0)\right] &\geqslant \mathbb{E} \left[ y ^2 \frac{x^2(n)}{2n^2}\xi^2_0\right] -\mathbb{E} \left[ y ^2 \frac{x^2(n)}{2n^2}\xi^2_0\mathbf{I}(\overline{ B }_0)\right]\nonumber\\
   &\geqslant  y ^2 \frac{x^2(n)}{2n^2}\sigma^2_0 - y ^2 \frac{x^2(n)}{2n^2} (\mathbb{E} [\xi^4_0])^{\frac{1}{2}}(\mathbb{V}(\overline{ B }_0))^{\frac{1}{2}}\nonumber\\
   &\geqslant  y ^2 \frac{x^2(n)}{2n^2}\sigma^2_0 - y ^2 \frac{x^2(n)}{2n^2} (\mathbb{E} [\xi^4_0])^{\frac{1}{2}}\left(\frac{K\mathbb{E} \left[e^{q\vert X_1\vert^{\alpha}}\right]}{e^{qx(n)^{\alpha}}}\right)^{\frac{1}{2}}.
 \end{align}
 \endgroup
 Now we get an upper bound for $\mathbb{E}_4$. By using moment condition (\ref{exp_moment}), Cauchy--Bunyakovsky--Schwarz inequality and Chebyshev inequality with the function $f(x) = e^{qx^{\alpha}}$ (see lemma \ref{geld_cheb}, (i) and (ii) respectively) we get the following bound
 \begingroup
 \allowdisplaybreaks
 \begin{align}\label{E4_upper}
   \mathbb{E}\left[ y  \frac{x(n)}{n}(-\xi_0)\mathbf{I}( B _0) + \mathbf{I}(\overline{ B }_0)\right] &= \mathbb{E} \left[ y \frac{x(n)}{n}(-\xi_0) +  y \frac{x(n)}{n}\xi_0\mathbf{I}(\overline{ B }_0)+ \mathbf{I}(\overline{ B }_0) \right]\nonumber\\
   &\leqslant \mathbb{E} \left[\vert  y \vert\frac{x(n)}{n}\vert \xi_0\vert\mathbf{I}(\overline{ B }_0)\right] +\mathbb{V}(\overline{ B }_0)\nonumber\\
   &\leqslant \left(\mathbb{E} \left[ y ^2 \frac{x^2(n)}{n^2}\xi_0^2\right]\right)^{\frac{1}{2}}\left(\frac{K\mathbb{E} [e^{q\vert X_1\vert^{\alpha}}]}{e^{q x^{\alpha}(n)}}\right)^{\frac{1}{2}} \nonumber\\
   &+ \frac{K\mathbb{E} [e^{q\vert X_1\vert^{\alpha}}]}{e^{q x^{\alpha}(n)}}\nonumber\\
   &= \left(  y ^2\frac{x^2(n)KM}{n^2e^{qx^{\alpha}(n)}}\sigma_0^2\right)^{\frac{1}{2}} + \frac{KM}{e^{q x^{\alpha}(n)}}.
 \end{align}
 \endgroup
 By using bounds (\ref{E3_lower}) and (\ref{E4_upper}) we get the following bound
 \begin{multline}\label{XI_lower}
   \mathbb{E} \left[e^{ y  \frac{x(n)}{n}\xi_0}\mathbf{I}( B _0)\right] \geqslant  y ^2 \frac{x^2(n)}{2n^2}\sigma^2_0 - y ^2 \frac{x^2(n)}{2n^2} (\mathbb{E} [\xi^4_0])^{\frac{1}{2}}\left(\frac{K\mathbb{E} \left[e^{q\vert X_1\vert^{\alpha}}\right]}{e^{qx(n)^{\alpha}}}\right)^{\frac{1}{2}}\\
   -\left(  y ^2\frac{x^2(n)KM}{n^2e^{qx^{\alpha}(n)}}\sigma_0^2\right)^{\frac{1}{2}} - \frac{KM}{e^{q x^{\alpha}(n)}} - \frac{\vert y \vert^3K^4 C^3x^3(n)}{3!n^3}M + 1\\
   =1+  y ^2 \frac{x^2(n)}{2n^2}\sigma^2_0 + o \left(\frac{x^2(n)}{n^2}\right),
 \end{multline}
 as $n\to\infty$.\par
 Using inequalities (\ref{XI_upper}) and (\ref{XI_lower}) we have
 \begin{multline}\label{eq4}
   \mathbb{E} \left[e^{ y  \frac{x(n)}{n}\xi_0}\mathbf{I}( B _0)\right]=1+  y ^2 \frac{x^2(n)}{2n^2}\sigma^2_0 + o \left(\frac{x^2(n)}{n^2}\right)\\
   = \exp\left\{\ln \left(1+  y ^2 \frac{x^2(n)}{2n^2}\sigma^2_0 + o \left(\frac{x^2(n)}{n^2}\right)\right)\right\}.
 \end{multline}
 Using the fact that $\ln(1+u) = u+o(u)$ as $u\to 0$, from (\ref{eq4}) we get
 \[
  \mathbb{E} \left[e^{ y  \frac{x(n)}{n}\xi_0}\mathbf{I}( B _0)\right]
  = \exp\left\{\frac{x^2(n)}{n^2}\left(\frac{\sigma^2_0 y ^2}{2}+o(1)\right)\right\},
 \]
 as $n\to\infty$, as was to be shown.\par
 Now we get an upper bound for
 \[
   \mathbb{E} \left[\mathbf{I}(\overline{C}_n)\prod\limits_{t=0}^{l-1}e^{y \frac{x(n)}{n}\xi_t}\mathbf{I}(B_t)\right].
 \]
By using Cauchy--Bunyakovsky--Schwarz inequality (see lemma \ref{geld_cheb}, (i)), the fact that $\xi_t$ is independent from $\xi_1,\ldots,\xi_{t-1}$, $1\leqslant t\leqslant l-1$, the fact that $\{X_i\}^{\infty}_{i=1}$ is strictly stationary, equality (\ref{eq3}) and the fact, that $\lim\limits_{n\to\infty}\frac{l}{n} = \frac{1}{K+m}$, we get
\begingroup
\allowdisplaybreaks
\begin{multline}\label{eq5}
  \mathbb{E} \left[\mathbf{I}(\overline{C}_n)\prod\limits_{t=0}^{l-1}e^{y \frac{x(n)}{n}\xi_t}\mathbf{I}(B_t)\right]\leqslant \left(\mathbb{E}[\mathbf{I}(\overline{C}_n)]\right)^{\frac{1}{2}}\left(\mathbb{E} \left[\prod\limits_{t=0}^{l-1}e^{2y \frac{x(n)}{n}\xi_t}\mathbf{I}(B_t)\right]\right)^{\frac{1}{2}}\\
  \leqslant \left(\sum\limits_{i=1}^{n}\mathbb{V}(\vert X_i\vert > x(n))\right)^{\frac{1}{2}}\left(\mathbb{E}\left[e^{2y \frac{x(n)}{n} \xi_0}\mathbf{I}(B_0)\right]\right)^{\frac{l}{2}}\\
  \leqslant \big(n\mathbb{V}(\vert X_1\vert > x(n))\big)^{\frac{1}{2}}\exp\left\{ l\frac{x^2(n)}{n^2}\left(\sigma^2_0y^2 + o(1)\right)\right\}\\
  \leqslant \big(n\mathbb{V}(\vert X_1\vert > x(n))\big)^{\frac{1}{2}}\exp\left\{ \frac{x^2(n)}{n}\left(\frac{\sigma^2_0}{K+m}y^2 + o(1)\right)\right\},\\
\end{multline}
\endgroup
as $n\to\infty$.
 By using Chebyshev inequality with the function $f(x)=e^{qx^{\alpha}(n)}$ (see lemma \ref{geld_cheb}, (ii)) and moment condition (\ref{exp_moment}), we get
\begin{equation}\label{eq6}
   \big(n\mathbb{V}(\vert X_1\vert > x(n))\big)^{\frac{1}{2}} \leqslant n^{\frac{1}{2}}M^{\frac{1}{2}}e^{-\frac{q}{2}x^{\alpha}(n)}.
\end{equation}
By combining inequalities (\ref{eq5}) and (\ref{eq6}), we get the following bound
\begin{multline}\label{second_subex}
 \mathbb{E} \left[\mathbf{I}(\overline{C}_n)\prod\limits_{t=0}^{l-1}e^{y \frac{x(n)}{n}\xi_t}\mathbf{I}(B_t)\right]\\
 \leqslant n^{\frac{1}{2}}M^{\frac{1}{2}}\exp\left\{\frac{x^2(n)}{n}\left(\frac{\sigma^2_0}{K+m}y^2 + o(1)\right) - \frac{q}{2}x^{\alpha}(n)\right\}.
\end{multline}
 \par
Using inequalities (\ref{eq1}), (\ref{eq3}) and the fact that $\lim\limits_{n\to\infty}\frac{l}{n} = \frac{1}{K+m}$, we have that
\begingroup
\allowdisplaybreaks
\begin{align*}
\mathbb{E} \left[e^{y \frac{x(n)}{n}I_1}\mathbf{I}(A_n)\right] &\leqslant \left(\mathbb{E}\left[e^{y \frac{x(n)}{n}\xi_0}\mathbf{I}(B_0)\right]\right)^{l}\\
&=\exp\left\{\frac{x^2(n)}{n^2}l \left(\frac{\sigma^2_0 y^2}{2} + o(1)\right)\right\}\\
&=\exp\left\{\frac{x^2(n)}{n} \frac{1}{K+m} \left(\frac{\sigma^2_0 y^2}{2} + o(1)\right)\right\},
\end{align*}
\endgroup
as $n\to\infty$. From the above inequality it follows that
\begin{equation}\label{eq_main_upper}
  \limsup\limits_{n\to\infty}\frac{n}{x^2(n)}\ln\mathbb{E} \left[e^{y \frac{x(n)}{n}I_1}\mathbf{I}(A_n)\right] \leqslant \frac{1}{K+m} \frac{\sigma^2_0y^2}{2}.
\end{equation}\par
By using inequalities (\ref{eq2}), (\ref{eq3}) and (\ref{second_subex}), we get
\begingroup
\allowdisplaybreaks
\begin{multline}\label{eq_lower}
  \mathbb{E} \left[e^{y \frac{x(n)}{n}I_1}\mathbf{I}(A_n)\right] \geqslant \exp\left\{\frac{x^2(n)}{n} \frac{1}{K+m} \left(\frac{\sigma^2_0 y^2}{2} + o(1)\right)\right\}\\
  -n^{\frac{1}{2}}M^{\frac{1}{2}}\exp\left\{\frac{x^2(n)}{n}\left(\frac{\sigma^2_0}{K+m}y^2 + o(1)\right) - \frac{q}{2}x^{\alpha}(n)\right\}.
\end{multline}
\endgroup
From inequality (\ref{eq_lower}) it follows that
\begingroup
\allowdisplaybreaks
\begin{multline*}
  \frac{n}{x^2(n)}\ln\mathbb{E} \left[e^{y \frac{x(n)}{n}I_1}\mathbf{I}(A_n)\right]\geqslant\\
  \begin{aligned}
    &\frac{n}{x^2(n)}\ln\left( e^{\frac{x^2(n)}{n} \frac{1}{K+m} \left(\frac{\sigma^2_0 y^2}{2} + o(1)\right)} -
    e^{\frac{\ln nM}{2} + \frac{x^2(n)}{n}\left(\frac{\sigma^2_0}{K+m}y^2 + o(1)\right) - \frac{q}{2}x^{\alpha}(n)}\right)\\
    &=  \frac{1}{K+m} \left(\frac{\sigma^2_0 y^2}{2} + o(1)\right)\\
    &+ \frac{n}{x^2(n)}\ln \left(1-e^{\frac{\ln nM}{2} + \frac{x^2(n)}{n}\left(\frac{\sigma^2_0}{K+m}y^2 + o(1)\right) - \frac{x^2(n)}{n} \frac{1}{K+m} \left(\frac{\sigma^2_0 y^2}{2} + o(1)\right) - \frac{q}{2}x^{\alpha}(n)} \right)\\
    &=  \frac{1}{K+m} \left(\frac{\sigma^2_0 y^2}{2} + o(1)\right)\\
    &+ \frac{n}{x^2(n)}\ln \left(1-e^{-x^{\alpha}(n)\left(-\frac{\ln nM}{2x^{\alpha}(n)} - \frac{x^{2-\alpha}}{n}\left(-\frac{\sigma^2_0}{K+m}y^2 + o(1)\right) + \frac{x^{2-\alpha}(n)}{n} \frac{1}{K+m} \left(\frac{\sigma^2_0 y^2}{2} + o(1)\right) + \frac{q}{2}\right)}\right),
  \end{aligned}
\end{multline*}
from which it follows that
\begin{equation}\label{eq_main_lower}
  \liminf\limits_{n\to\infty}\frac{n}{x^2(n)}\ln\mathbb{E} \left[e^{y \frac{x(n)}{n}I_1}\mathbf{I}(A_n)\right]\geqslant \frac{1}{K+m} \frac{\sigma^2_0 y^2}{2}.
\end{equation}
Proposition 1 follows from inequalities (\ref{eq_main_upper}) and (\ref{eq_main_lower}).
\endgroup
  \par\noindent
  \textbf{Proposition 2.} For any fixed $y\in\mathbb{R}$, $K\geqslant m+1$ the following inequality holds
  \[
    \lim\limits_{n\to\infty} \frac{n}{x^2(n)}\ln\mathbb{E} \left[e^{y \frac{x(n)}{n}I_2}\mathbf{I}(A_n)\right] \leqslant \frac{1}{K+m}\frac{y^2\widetilde{\sigma}^2_0}{2},
  \]
  where $\widetilde{\sigma}^2_0 = \mathbb{E} \left[\eta^2_0\right]$.\par\noindent
  \textit{Proof of proposition 2}. The proof is completely parallel to that of proposition 1, thus we omit it.\par\noindent
  \textbf{Proposition 3.} For any fixed $y\in\mathbb{R}$, $K\geqslant m+1$ the following inequality holds
  \[
    \limsup\limits_{n\to\infty} \frac{n}{x^2(n)}\ln\mathbb{E}\left[e^{y \frac{x(n)}{n}I_3}\mathbf{I}(A_n)\right]\leqslant 0.
  \]\par\noindent
  \textit{Proof of proposition 3}.
  By using lemma \ref{liu_lemma} and the fact that $\{X_i\}_{i=1}^{\infty}$ is strictly stationary, we get the following bound
  \begingroup
  \allowdisplaybreaks
  \begin{multline*}\label{J_1_upper}
    \frac{n}{x^2(n)}\ln\mathbb{E}\left[e^{y \frac{x(n)}{n}I_3}\mathbf{I}(A_n)\right]\\
    \begin{aligned}
       &\leqslant\frac{n}{x^2(n)}\ln\mathbb{E} \left[\exp \left\{\vert y \vert   \frac{x(n)}{n} \sum\limits_{i=l(K+m)+1}^{n}\vert X_i\vert\right\}\prod\limits_{i=l(K+m)+1}^{n}\mathbf{I}(\vert X_i\vert \leqslant x(n))\right]\\
    &=   \frac{n}{x^2(n)}\ln\mathbb{E} \left[\exp \left\{\vert y \vert   \frac{x(n)}{n} \sum\limits_{i=1}^{n-l(K+m)}\vert X_i\vert\right\}\prod\limits_{i=1}^{n-l(K+m)}\mathbf{I}(\vert X_i\vert \leqslant x(n))\right]\\
    &\leqslant   \frac{n}{x^2(n)}\ln\mathbb{E} \left[\exp \left\{\vert y \vert   \frac{x(n)}{n} \sum\limits_{i=1}^{n-l(K+m)}\vert X_i\vert\mathbf{I}(\vert X_i\vert \leqslant x(n))\right\}\right]\\
    &\leqslant    \frac{n}{x^2(n)}\ln\mathbb{E} \left[\exp\left\{\vert y \vert   \frac{x(n)}{n} (K+m)\sum\limits_{i=1}^{K+m}\frac{\vert X_i\vert\mathbf{I}(\vert X_i\vert \leqslant x(n))}{K+m}\right\}\right] \\
    &\leqslant    \frac{n}{x^2(n)}\sum\limits_{i=1}^{K+m}\left(\frac{1}{K+m}\ln\mathbb{E}\left[\exp\left\{\vert y \vert   \frac{x(n)}{n}(K+m)\vert X_i\vert \mathbf{I}(\vert X_i\vert \leqslant x(n))\right\}\right]\right) \\
    &=   \frac{n}{x^2(n)}\ln\mathbb{E}\left[\exp\left\{\vert y \vert   \frac{x(n)}{n}(K+m)\vert X_1\vert \mathbf{I}(\vert X_1\vert \leqslant x(n))\right\}\right]\\
    &=   \frac{n}{x^2(n)}\ln\mathbb{E}\left[\exp\left\{\vert y \vert   \frac{x(n)^{2-\alpha}}{n}(K+m)\vert X_1\vert^{\alpha} \mathbf{I}(\vert X_1\vert \leqslant x(n))\right\}\right]\\
    \end{aligned}\\
    \leqslant    \frac{n}{x^2(n)}\ln\mathbb{E}\left[\exp\left\{\vert y \vert   \frac{x(n)^{2-\alpha}}{n}(K+m)\vert X_1\vert^{\alpha}\right\}\right].
  \end{multline*}
  \endgroup
  Using condition (\ref{seq}) for large enough $n$ we have
  \[
    \vert y\vert \frac{x(n)^{2-\alpha}}{n}(K+m) \leqslant q.
  \]
  By using the above inequality and moment condition (\ref{exp_moment}), for large enough $n$ we get
  \begin{equation*}
    \frac{n}{x^2(n)}\ln\mathbb{E}\left[e^{y \frac{x(n)}{n}I_3}\mathbf{I}(A_n)\right] \leqslant \frac{n}{x^2(n)}\ln M,
  \end{equation*}
  from which it follows that
  \begin{equation}\label{J_1_upper}
    \limsup\limits_{n\to\infty}\frac{n}{x^2(n)}\ln\mathbb{E}\left[e^{y \frac{x(n)}{n}I_3}\mathbf{I}(A_n)\right] \leqslant 0.
  \end{equation}
  Proposition 3 is proven.
\par
  \par
  Now we proceed with the proof of lemma \ref{eq_lemma}. Fix $p_1 > 1, p_2 > 1, q_1 > 1, q_2 > 1$ such that $\frac{1}{p_1} + \frac{1}{q_1} = 1$, $\frac{1}{p_2} + \frac{1}{q_2} = 1$. By using H\"{o}lder inequality (see lemma \ref{geld_cheb}, (i)) for any fixed $\lambda \in \mathbb{R}$ we get
  \begingroup
  \allowdisplaybreaks
  \begin{multline}\label{eq8}
    \frac{n}{x^2(n)}\ln\mathbb{E} \left[e^{\lambda \frac{x(n)}{n}S_n}\mathbf{I}(A_n)\right]  \\
    \leqslant \frac{1}{q_1}  \frac{n}{x^2(n)}\ln\mathbb{E} \left[e^{\lambda q_1 \frac{x(n)}{n}I_3}\mathbf{I}(A_n)\right] + \frac{1}{p_1}  \frac{n}{x^2(n)}\ln\mathbb{E}\left[e^{\lambda p_1 \frac{x(n)}{n}(I_1 + I_2)}\mathbf{I}(A_n)\right]\\
    \leqslant \frac{1}{q_1}  \frac{n}{x^2(n)}\ln\mathbb{E}\left[e^{\lambda q_1 \frac{x(n)}{n}I_3}\mathbf{I}(A_n)\right] + \frac{1}{p_1p_2}  \frac{n}{x^2(n)}\ln\mathbb{E} \left[e^{\lambda p_1p_2 \frac{x(n)}{n}I_1}\mathbf{I}(A_n)\right]\\
    + \frac{1}{p_1q_2}  \frac{n}{x^2(n)}\ln \mathbb{E}\left[e^{\lambda p_1q_2 \frac{x(n)}{n}I_2}\mathbf{I}(A_n)\right].
  \end{multline}\par
  \endgroup
  By using propositions 1--3 and inequality (\ref{eq8}), we get
  \begin{multline*}
    \limsup\limits_{n\to\infty}\frac{n}{x^2(n)}\ln\mathbb{E} \left[e^{\lambda \frac{x(n)}{n}S_n}\mathbf{I}(A_n)\right] \leqslant p_1p_2\frac{1}{K+m}\frac{\sigma^2_0\lambda^2}{2} + \frac{p_1q_2}{K+m}\frac{\widetilde{\sigma}^2_0\lambda^2}{2}\\
    = p_1p_2\frac{K}{K+m} \frac{\lambda^2}{2}\frac{1}{K}\mathbb{E} \left[\sum\limits_{i=1}^{K}X_i \right]^2 + \frac{p_1q_2}{K+m}\frac{\lambda^2}{2}\mathbb{E} \left[\sum\limits_{i=1}^{m}X_i \right]^2.
  \end{multline*}
  Since the left-hand side of the above inequality does not depend on $K$, by letting $K\to\infty$, we get
  \[
    \limsup\limits_{n\to\infty}\frac{n}{x^2(n)}\ln\mathbb{E} \left[e^{\lambda \frac{x(n)}{n}S_n}\mathbf{I}(A_n)\right] \leqslant p_1p_2 \frac{\sigma^2\lambda^2}{2}.
  \]
  Since the left-hand side of the above inequality does not depend on $p_1$, $p_1$, $q_1$ and $q_2$, by letting $p_1\to 1$ and $p_2\to 1$, we get
  \begin{equation}\label{main_upper}
    \limsup\limits_{n\to\infty}\frac{n}{x^2(n)}\ln\mathbb{E} \left[e^{\lambda \frac{x(n)}{n}S_n}\mathbf{I}(A_n)\right] \leqslant \frac{\sigma^2\lambda^2}{2}.
  \end{equation}\par
  Now we find a lower bound for
  \[
      \liminf\limits_{n\to\infty}\frac{n}{x^2(n)}\ln\mathbb{E} \left[e^{\lambda \frac{x(n)}{n}S_n}\mathbf{I}(A_n)\right].
  \]
  Using H\"{o}lder inequality (see lemma \ref{geld_cheb}, (i)), we have that
    \begingroup
    \allowdisplaybreaks
    \begin{multline*}
      \ln\mathbb{E}\left[e^{\lambda \frac{x(n)}{p_1p_2n}I_1}\mathbf{I}(A_{n})\right]  = \ln\mathbb{E}\left[e^{\lambda \frac{x(n)}{p_1p_2n}(I_1 + I_2 - I_2)}\mathbf{I}(A_{n})\right]\\
      \leqslant \frac{1}{p_1}\ln\mathbb{E}\left[e^{\lambda \frac{x(n)}{p_2n}(I_1 + I_2)}\mathbf{I}(A_{n})\right] + \frac{1}{q_1}\ln\mathbb{E}\left[e^{-\lambda \frac{q_1x(n)}{p_1p_2n}I_2}\mathbf{I}(A_{n})\right]\\
      \leqslant \frac{1}{p_1}\ln\mathbb{E}\left[e^{\lambda \frac{x(n)}{p_2n}(I_1 + I_2 + I_3 - I_3)}\mathbf{I}(A_{n})\right] + \frac{1}{q_1}\ln\mathbb{E}\left[e^{-\lambda \frac{q_1x(n)}{p_1p_2n}I_2}\mathbf{I}(A_{n})\right]\\
      \leqslant \frac{1}{p_1p_2}\ln\mathbb{E}\left[e^{\lambda \frac{x(n)}{n}(I_1 + I_2 + I_3)}\mathbf{I}(A_{n})\right] + \frac{1}{p_1q_2} \ln\mathbb{E}\left[e^{-\lambda \frac{q_2x(n)}{p_2n}I_3}\mathbf{I}(A_{n})\right]\\
      +\frac{1}{q_1}\ln\mathbb{E}\left[e^{-\lambda \frac{q_1x(n)}{p_1p_2n}I_2}\mathbf{I}(A_{n})\right].
    \end{multline*}
    \endgroup
    By using the above we get
    \begingroup
    \allowdisplaybreaks
    \begin{multline}\label{eq9}
      \frac{n}{x^2(n)}\ln\mathbb{E}\left[e^{\lambda \frac{x(n)}{n}S_{n}}\mathbf{I}(A_{n})\right] \geqslant - \frac{p_2}{q_2} \frac{n}{x^2(n)} \ln\mathbb{E}\left[e^{-\lambda \frac{q_2x(n)}{p_2n}I_3}\mathbf{I}(A_{n})\right]\\
      +p_1p_2 \frac{n}{x^2(n)}\ln\mathbb{E}\left[e^{\lambda \frac{x(n)}{p_1p_2n}I_1}\mathbf{I}(A_{n})\right]     - \frac{p_1p_2}{q_1} \frac{n}{x^2(n)}\ln\mathbb{E}\left[e^{-\lambda \frac{q_1x(n)}{p_1p_2n}I_2}\mathbf{I}(A_{n})\right].
    \end{multline}
    \endgroup
    By using propositions 1-3 and inequality (\ref{eq9}), we get
    \begingroup
    \allowdisplaybreaks
    \begin{multline*}
      \liminf\limits_{n\to\infty}\frac{n}{x^2(n)}\ln\mathbb{E}\left[e^{\lambda \frac{x(n)}{n}S_{n}}\mathbf{I}(A_{n})\right]\\
      \geqslant  \frac{1}{p_1p_2}\frac{K}{K+m} \frac{\lambda^2}{2}\frac{1}{K}\mathbb{E} \left[\sum\limits_{i=1}^{K}X_i \right]^2 - \frac{q_1}{p_1p_2}\frac{1}{K+m}\frac{\lambda^2}{2}\mathbb{E} \left[\sum\limits_{i=1}^{m}X_i \right]^2.
    \end{multline*}
    \endgroup
    Since the left-hand side of the above inequality does not depend on $K$, by letting $K\to\infty$, we get
    \[
      \liminf\limits_{n\to\infty}\frac{n}{x^2(n)}\ln\mathbb{E} \left[e^{\lambda \frac{x(n)}{n}S_n}\mathbf{I}(A_n)\right] \geqslant \frac{1}{p_1p_2} \frac{\sigma^2\lambda^2}{2}.
    \]
    Since the left-hand side of the above inequality does not depend on $p_1$, $p_1$, $q_1$ and $q_2$, by letting $p_1\to 1$ and $p_2\to 1$, we get
    \begin{equation}\label{main_lower}
      \liminf\limits_{n\to\infty}\frac{n}{x^2(n)}\ln\mathbb{E} \left[e^{\lambda \frac{x(n)}{n}S_n}\mathbf{I}(A_n)\right] \geqslant \frac{\sigma^2\lambda^2}{2}.
    \end{equation}\par
  By combining (\ref{main_upper}) and (\ref{main_lower}), we get
  \[
    \lim\limits_{n\to\infty} \frac{n}{x^2(n)}\ln\mathbb{E}\left[e^{\lambda \frac{x(n)}{n}S_n}\mathbf{I}(A_n)\right]= \frac{\sigma^2\lambda^2}{2}.
  \]
\end{proof}

The authors are grateful to A.I. Sakhanenko and  reviewer for valuable comments that improved the manuscript.

\bigskip

\end{document}
